Author’s response to reviews

Title: Process evaluation of appreciative inquiry to translate pain management evidence into pediatric nursing practice

Authors:

Tricia Kavanagh (tricia.orr@utoronto.ca)
Bonnie Stevens (b.stevens@utoronto.ca)
Kate Seers (kate.seers@warwick.ac.uk)
Souraya Sidani (ssidani@ryerson.ca)
Judy Watt-Watson (j.watt.watson@utoronto.ca)

Version: 3 Date: 4 November 2010

Author’s response to reviews: see over
November 2, 2010

Dear Drs. Eccles and Mittman;

Please find attached our resubmission of the manuscript entitled ‘Process evaluation of appreciative inquiry to translate pain management evidence into pediatric nursing practice’ for publication in *Implementation Science*. We have accepted most of the changes made by the copy editor, with the exception of the following: (a) ‘conducted in’ (p. 2) was deleted, as the unit was the case; and (b) ‘those’ was changed to ‘that’ (p. 25), as only one context was considered in this study. All comments were addressed, including: (a) providing a title and legend for Figure 1; (b) correcting the spelling of ‘NVivo 8’ (p. 9); and (c) spelling out ‘IQR’ in the notes for Table 3 (p. 36).

Please do not hesitate to contact me by email at tricia.orr@utoronto.ca if you have any questions regarding this resubmission.

Sincerely,

Tricia Kavanagh, RN, PhD (on behalf of authors)